Author's response to reviews

Title: A proposed core curriculum for dental English education in Japan

Authors:

Omar M Rodis (omarodis@tokushima-u.ac.jp)
Edward Barroga (eb@dimc-tmu.jp)
Patrick Barron (jpatrickbarron@gmail.com)
James Hobbs (jahobbs@iwate-med.ac.jp)
Asiri Jayawardena (asiri@tsurumi-u.ac.jp)
Ikuo Kageyama (kageyama@ngt.ndu.ac.jp)
Bukasa Kalubi (kalubi@basic.med.tokushima-u.ac.jp)
Clive Langham (clive25@hotmail.com)
Yoshizo Matsuka (matsuka@tokushima-u.ac.jp)
Yoichiro Miyake (miyake@tokushima-u.ac.jp)
Naoko Seki (nseki.edev@tmd.ac.jp)
Hiroko Oka (okhiroko@hiroshima-u.ac.jp)
Martin Peters (martin1948@mac.com)
Yo Shibata (yookun@dent.showa-u.ac.jp)
Roxana Stegaroiu (roxana@dent.niigata-u.ac.jp)
Kazuyoshi Suzuki (kazukiti@dpc.agu.ac.jp)
Shigeru Takahashi (tshigeru@den.hokudai.ac.jp)
Hironori Tsuchiya (hiro@dent.asahi-u.ac.jp)
Toshiko Yoshida (toshiko@md.okayama-u.ac.jp)
Katsuhiko Yoshimoto (yoshimoto@tokushima-u.ac.jp)

Version: 3
Date: 21 April 2014

Author's response to reviews:

April 21, 2014

Dr. Fernando Marques
Executive Editor
BMC Medical Education

Dear Dr. Marques,

On behalf of my co-authors, I would like to resubmit our manuscript entitled “A Proposed Core Curriculum for Dental English Education in Japan” for publication in BMC Medical Education. The manuscript is part of a full report (available as a supplement file) containing results of two discussion forums and a workshop/seminar. The project was supported by a 5-million yen grant (No. 23531201) for Scientific Research (C) (General) (2011-2014) entitled “Developing a Core Curriculum for the Dental English Course for Dental Schools...
in Japan” from the Japan Ministry of Education and the Japan Society for the Promotion of Science.

We have modified the manuscript as instructed:
1. Reformat: We reformatted our manuscript as a research article.
2. Abstract: We have included the Abstract as part of the manuscript.

Thank you very much for considering our manuscript.

Respectfully yours,

Omar M.M. Rodis